Author's response to reviews

Title: Impact of the joint association between sex, age and diabetes on long-term mortality after acute myocardial infarction

Authors:

Fei Gao (gao.fei@nhcs.com.sg)
Carolyn SP Lam (carolyn_lam@nuhs.edu.sg)
Ling Ling Sim (sim.ling.ling@nhcs.com.sg)
Tian Hai Koh (koh.tian.hai@nhcs.com.sg)
David Foo (david_foo@ttsh.com.sg)
Hean Yee Ong (ong.hean.yee@alexandrahealth.com.sg)
Khim Leng Tong (khim_leng_tong@cg.com.sg)
Huay Cheem Tan (huay_cheem_tan@nuhs.edu.sg)
David Machin (dm113@leicester.ac.uk)
Kok Seng Wong (wong.kok.seng@sgh.com.sg)
Mark YY Chan (mark_chan@nuhs.edu.sg)
Terrance SJ Chua (terrance.chua.s.j@nhcs.com.sg)

Version: 3 Date: 28 January 2015

Author’s response to reviews:

Manuscript revised to include specific names of the ethics committees (Page 7 Line 7-8).